Author's response to reviews

Title: Changes in the flexion-relaxation response induced by hip extensor and erector spinae muscle fatigue.

Authors:

Martin Descarreaux (martin.descarreaux@uqtr.ca)
Danik Lafond (danik.lafond@uqtr.ca)
Vincent Cantin (vincent.cantin@uqtr.ca)

Version: 4  Date: 28 May 2010

Author's response to reviews:

Dear editor,

here is a revised version of our manuscript where all the formatting changes (major and minor) have been addressed. Title page, affiliation, abstract and figure legend have all been modified according to the recommendations provided.

Sincerely,

Martin Descarreaux